Author's response to reviews

Title: A randomized clinical trial of the effectiveness of mechanical traction for sub-groups of patients with low back pain: study methods and rationale

Authors:

Julie M Fritz (julie.fritz@hsc.utah.edu)
Anne Thackeray (A.Thackeray@m.cc.utah.edu)
John D Childs (john@texaspts.com)
Gerard P Brennan (Gerard.Brennan@imail.org)

Version: 3 Date: 28 April 2010

Author's response to reviews: see over
We have made the following modifications in response to the revisions requested:

-------------

Major revisions (we require the author to make these changes)

References: Please make sure all references are cited in numerical order. Some of your reference citations are out of sequence.

We double checked the references and corrected those that were out of numerical order.

Tables: we are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold.

We revised the tables to remove colors and shading.

The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key) below the table.

We revised the table titles as requested.

Please remove the visible vertical lines from your tables.

We revised the tables to remove vertical lines.

Manuscript: this manuscript does not have the correct structure. Please reformat it to use these headings:

Abstract; Background; Methods/Design; Discussion; Abbreviations (if any); Competing interests; Authors' contributions; Authors? information (optional); Acknowledgements (optional); References

We reformatted the manuscript using the requested headings.

---------------

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, ?The Title of My Prestigious Publication: A Scientific Study.? should be changed to ?The title of my prestigious publication: a scientific study?. Remember to update the submission system to make sure the title remains consistent.

We reformatted the title as requested.

Email address: For the authors' email address list, please use the following format as an example:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

We reformatted the e-mail addresses as requested.

References: Please ensure all citations are placed before the full-stop.
References: the reference list should contain up to 30 authors’ names for each citation. The term “et al.” should not be used if there are fewer than 30 authors.

We reformatted the references and citations as requested.